**Author's response to reviews**

**Title:** Mental Illness is Common With Functional Gastrointestinal Disorder (FGID) and is Important to Consider At Patient Consultation

**Authors:**

- Ture Aalander (ture@tapraktik.se)
- Kurt Svaerdsudd (Kurt.Svardsudd@pubcare.uu.se)
- Sven-Erik Johansson (lmse.johansson@swipnet.se)
- Lars Agreus (lars.agreus@klinvet.ki.se)

**Version:** 2  **Date:** 1 October 2004

**Author's response to reviews:**

The list of abbreviations used is made into alphabetical order. The Author's contributions is added. The reference no 53 (Sackett DL) is added. Spaces are taken away between reference numbers. Formatting of the legends of the figures and tables with arabic numerals and titles. The Medical knowledge questionnaire, translated from Swedish, is added to the Appendix.